Author's response to reviews

Title: Logistic Organ Dysfunction Score (LODS): A reliable postoperative risk management score also in cardiac surgical patients?

Authors:

Matthias B Heldwein (mathias.heldwein@med.uni-jena.de)
Akmal M.A. Badreldin (akmalbadreldin@yahoo.com)
Fabian Doerr (fabidoe@web.de)
Thomas Lehmann (thomas.lehmann@mti.uni-jena.de)
Ole Bayer (ole.bayer@med.uni-jena.de)
Torsten Doenst (ht@med.uni-jena.de)
Khosro Hekmat (hekmat@med.uni-jena.de)

Version: 2 Date: 1 August 2011

Author's response to reviews:

Dear Sirs,

Thank you very much for your valuable review. We have considered the comments of all three reviewers and integrated them very carefully in our manuscript.

The new revised version has all changes in red.

We hope that find our revision acceptable. If you find the manuscript still wanting in any details, please do not hesitate to contact us.

A. Badreldin and co-authors